Author's response to reviews

Title: Hem-O-Lok clip: a neglected cause of severe bladder neck contracture and consequent urinary incontinence after robot-assisted laparoscopic radical prostatectomy.

Authors:

- Luigi Cormio (luigicormio@libero.it)
- Paolo Massenio (p.massenio@alice.it)
- Giuseppe Di Fino (giuseppe_difino@libero.it)
- Oscar Selvaggio (oscarsel@libero.it)
- Salvatore Micali (micali.salvatore@unimore.it)
- Giuseppe Carrieri (giuseppe.carrieri@unifg.it)

Version: 3
Date: 4 February 2014

Author's response to reviews: see over
To Hayley Henderson, MD
Senior Executive Editor
BMC Urology

Foggia, 03.02.2014

Dear Dr. Henderson,

Thank you very much for having revised our manuscript titled "Hem-O-Lok clip: a neglected cause of severe bladder neck contracture and consequent urinary incontinence after robot-assisted laparoscopic radical prostatectomy” and having given us the possibility of further improving it.

We noticed that the Reviewers did not suggest any change. As for the Editorial comments:

1. we revised the Figures and submitted them as single composite illustration files;
2. we formatted the manuscript according to the Journal style.

We sincerely this new version of our manuscript is now suitable for publication in your Journal and look forward to hearing from you at your soonest convenience.

Sincerely,

Luigi Cormio, MD PhD
on behalf of all authors